Author's response to reviews

Title: Growth patterns among HIV-exposed infants receiving nevirapine prophylaxis in Pune, India

Authors:

   Malathi Ram (mram@jhsph.edu)
   Nikhil Gupte (Nikhil_jhumit@yahoo.com)
   Uma Nayak (un8x@cms.mail.virginia.edu)
   Aarti A Kinikar (aarti.kinikar63@gmail.com)
   Mangesh Khandve (drmangesh@jhumitpune.com)
   Anita V Shankar (avshanka@jhsph.edu)
   Jayagowri Sastry (gowsas@gmail.com)
   Robert C Bollinger (rcb@jhmi.edu)
   Amita Gupta (agupta25@jhmi.edu)

Version: 3 Date: 9 April 2012

Author's response to reviews:

The word "undernourished" has been replaced by the word "underweight" in the manuscript.